Author's response to reviews

Title: Reproducibility of different screening classifications in ultrasonography of the newborn hip

Authors:

Christian D Peterlein (peterlei@med.uni-marburg.de)
Karl F Schüttler (Kalle-Schuettler@T-Online.de)
Stefan Lakemeier (lakemeie@med.uni-marburg.de)
Christian Görg (goergc@med.uni-marburg.de)
Nina Timmesfeld (timmesfe@med.uni-marburg.de)
Susanne Fuchs-Winkelmann (fuchss@med.uni-marburg.de)
Markus D Schofer (schofer@med.uni-marburg.de)

Version: 4 Date: 22 December 2010

Author's response to reviews: see over
Dear Editors,

Thank you for your thorough reading of the revised manuscript.

According to your instructions we have adjusted the manuscript:

- Title and author list match
- Tables are included as additional files
- Reference 33 now appears in the text
- Full citation details are provided for reference 21
- All highlighting and tracking are removed

We are looking forward to hearing from you.

Greetings from Marburg

Christian-Dominik Peterlein, MD
(Corresponding Author)